Author's response to reviews

Title: The effectiveness of case management for comorbid diabetes type 2 patients; the CasCo study. Design of a randomized controlled trial.

Authors:

Nathalie Versnel (n.versnel@nivel.nl)
Laura MC Welschen (l.welschen@vumc.nl)
Caroline A Baan (caroline.baan@rivm.nl)
Giel Nijpels (g.nijpels@vumc.nl)
François G Schellevis (f.schellevis@nivel.nl)

Version: 4 Date: 5 July 2011

Author's response to reviews: see over
Dear Editor,

In reply to your email (July 5th) we made the following changes in order to meet the journal style as requested.

- References in main text: are put in numerical order.
- Tables: merged cells are deleted.

I hope all necessary alterations are made and we are looking forward to the acceptance of the manuscript.

On behalf of all authors,
Yours sincerely,

Nathalie Versnel